Author's response to reviews

Title: Prevalence and sociodemographic correlates of stunting, underweight, and overweight among Palestinian school adolescents (13-15 years) in two major governorates in the West Bank

Authors:

Nahed Mikki (nahedmikki@yahoo.com)
Hanan F Abdul-Rahim (hanan.arahim@qu.edu.qa)
Faisal Awartani (faisal@alpha.ps)
Gerd Holmboe-Ottesen (gerd.holmboe-ottesen@medisin.uio.no)

Version: 8 Date: 23 November 2009

Author's response to reviews: see over
November 23rd, 2009.

Dear editor:
Thank you for accepting our manuscript for publication in the BMC Public Health. We have revised the manuscript according to the requested revisions as follows:

1. Tables: the tables have been included in the manuscript in a tables section following the references and have been cited in the manuscript as such. They have been also renumbered in an ascending numerical order: table 3a has been renumbered as table 3, table 3b has been renumbered as table 4 and table 4 has been renumbered as table 5. Changes in the text were made to correspond with the changes in the numbering of the tables.
2. Title page heading: this was removed from the title page
3. Key words: these were removed from the title page.
4. Typography: some typographical errors were found in the manuscript and in the tables and these were corrected.

The authors.